Author’s response to reviews

Title: Psychosocial Correlates with Depressive Symptoms Six years after a First Episode of Psychosis as Compared with Findings from a General Population Sample.

Authors:

Yvonne Forsell (Yvonne.Forsell@phs.ki.se)
Johan Cullberg (j.cullberg@swipnet.se)
Sonja Levander (Sonja.Levander@smd.sll.se)

Version: 2 Date: 17 September 2004

Author’s response to reviews:

Thank you for the valuable comments from the reviewers that we have taken into account in the following ways:

Reviewer 1.
A further explanation about the procedure with the questionnaire is now added in the method section. A reference to the construction of the questionnaire is added. See method section page 6, second paragraph.

Reviewer 2.
Most certainly we ought to have described the self-rating depression scale further and this is now added in the background section. The validity and reliability has been found to be good in previous studies. References to these studies are added. See background section page 3, third paragraph.